Author's response to reviews

Title: Did the pandemic have an impact on influenza vaccination attitude? A survey among health care workers

Authors:

Bilgin Arda (bilgin.arda@ege.edu.tr)
Raika Durusoy (raika.durusoy@ege.edu.tr)
Tansu Yamazhan (tansu.yamazhan@ege.edu.tr)
Oguz R Sipahi (oguz.resat.sipahi@ege.edu.tr)
Meltem Tasbakan (tasbakan@yahoo.com.tr)
Hüsnü Pullukcu (husnup@yahoo.com.tr)
Esra Erdem (esra.erdem@yahoo.com)
Sercan Ulusoy (sercan.ulusoy@ege.edu.tr)

Version: 7 Date: 23 March 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team Member,

All the required formatting changes were made. The detailed responses are below. Responses to the concerns are the ‘OK’s in bold. Yours,

Raika Durusoy

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs)

OK

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

OK

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Please remove all tracking.

OK

All text should be in black

OK

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance

OK